Author's response to reviews

Title: Escherichia coli strains of phylogenetic group B2 and D and bacteriocin production are associated with advanced colorectal neoplasia

Authors:

Darina Kohoutova (Darina.Kohoutova@seznam.cz)
David Smajs (dsmajs@med.muni.cz)
Paula Moravkova (paula.moravkova@fnhk.cz)
Jiri Cyrany (jiri.cyrany@email.cz)
Monika Moravkova (moravmon@seznam.cz)
Miroslava Forstlova (forstlova@fnhk.cz)
Michal Cihak (mcihak@post.cz)
Jan Bures (Bures@lfhk.cuni.cz)

Version: 3 Date: 4 November 2014

Author's response to reviews: see over
Dear Sirs,

Thank you for your kind comments to our article "Escherichia coli strains of phylogenetic group B2 and D and bacteriocin production are associated with advanced colorectal neoplasia".

We have made corrections according to the requirements / recommendations of the reviewer and we send our new version of the manuscript. Changes made in the manuscript are highlighted with the coloured (red) text.

We added “Supplementary material 2“ into the main text of the manuscript as “Table 1“ and “Table 2“, but we did not include „Supplementary material 1“ into the main text because of its magnitude.

Kind regards,

Darina Kohoutova, MD, Ph.D.